Author's response to reviews

Title: Periodontitis associated with Plasminogen Deficiency: A Case Report

Authors:

Sarah Neering (sarah.neering@med.uni-duesseldorf.de)
Sabine Adyani-Fard (sabine.adyani-fard@med.uni-duesseldorf.de)
Stefan Ruettermann (stefan.ruettermann@med.uni-duesseldorf.de)
Thomas F Flemmig (flemmig@uw.edu)
Thomas Beikler (beikler@med.uni-duesseldorf.de)

Version: 3  Date: 29 December 2014

Author's response to reviews:

Dear Eloisa Nolasco or to whom it may concern,

we have made the following changes to the manuscript:
1. title page now includes email addresses of all authors.
2. page, line numbering and list of abbreviations was added.
3. the sections "competing interests", "authors' contributions", "consent" and figure legends were added.

Further the manuscript is now provided as a word document and a Care checklist has been filled in.

As we present a case report and not a research on a human being, no approval of the ethics committee of the Heinrich-Heine-University was necessary.

I hope that the manuscript is now in order for peer reviewing.

Yours sincerely,

Sarah Neering